Author's response to reviews

Title: The Effect of an Educational Intervention on Coronary Artery Bypass Graft Surgery Patients' Participation Rate in Cardiac Rehabilitation Programs: A controlled Health Care Trial

Authors:

Rachel Dankner (racheld@gertner.health.gov.il)
Galit Geulayov (Galit.Geulayov@bristol.ac.uk)
Arnona Ziv (Arnonaz@gertner.health.gov.il)
Illia Novikov (IliaN@gertner.health.gov.il)
Uri Goldbourt (goldbu1@post.tau.ac.il)
Yaakov Drory (drory.yaacov@gmail.com)

Version: 2 Date: 20 August 2011

Author's response to reviews: see over
Dear Dr. Shipley,

I am pleased to submit a revised version of MS: 3182612845585254. Below is the list of changes made according to your instruction. These changes can be seen in the attached manuscript via the "track changes" mode.

1. The title of the manuscript was revised to better reflect the methodology of the study to: 
   "The Effect of an Educational Intervention on Coronary Artery Bypass Graft Surgery Patients' Participation Rate in Cardiac Rehabilitation Programs: A controlled Health Care Trial".
2. The abstract includes 270 words and is structured into separate sections headed Background, Methods/Design, Discussion.
3. No abbreviations or references are used in the abstract.
4. Trial Registration is listed in the abstract, along with the unique identifying number.
5. 'Authors' contributions' section was placed before the Reference list.
6. 'Competing interests' section was placed between the Conclusions and Authors' contributions section.

We hope that our manuscript in its revised form will merit publication in your journal.

Sincerely yours,

Dr. Timothy Shipley PhD

Tel: +44 (0) 20 3192 2013

e-mail: editorial@biomedcentral.com ; Web: http://www.biomedcentral.com/
Rachel Dankner, MD, MPH
Unit for Cardiovascular Epidemiology
Affiliated to the Tel-Aviv University
Sackler Faculty of Medicine

Fax. 5349607–03, Tel. 53–0305173
e-mail: racheld@gertner.health.gov.il